# How do Firms Issue Shares? Evidence from Around the World<sup> $\Psi$ </sup>

R. David McLean University of Alberta rdmclean@ualberta.ca

Tianyu Zhang City University of Hong Kong tyzhang@cityu.edu.hk

Mengxin Zhao University of Alberta mengxin.zhao@ualberta.ca

November 2010

## Abstract

We study how seasoned shares are issued in 41 countries during the period 1990-2008. We obtain issues data from SDC Thomson and manual Factiva searches, which increase the SDC sample by 43%. Public offerings are the most frequent type of share issue, while mergers are on average the largest. All types of issues increase in frequency during our sample period. Strong investor protection laws are associated with more frequent public offerings and mergers, and fewer rights offerings. The relation between investor protection and private placements is positive, but less robust as compared to other types of issues. We relate these findings to several theories of issuance choice. Stock market development has no effect on seasoned issuance frequency or type. The findings are robust to excluding U.S. firms.

 $<sup>^{\</sup>Psi}$  We thank Mark Huson, Espen Eckbo, Andrew Karoyli, Ron Masulis, Randall Morck, Jeffrey Pontiff, and seminar participants at the University of Alberta and the Shanghai University of Finance and Economics for helpful comments and discussions. Shansan Wang provided excellent research assistance. We are grateful for financial support from the City University of Hong Kong's Strategic Research Grant and the Social Sciences and Humanities Research Council of Canada.

# How do Firms Issue Shares? Evidence from Around the World

November 2010

# Abstract

We study how seasoned shares are issued in 41 countries during the period 1990-2008. We obtain issues data from SDC Thomson and manual Factiva searches, which increase the SDC sample by 43%. Public offerings are the most frequent type of share issue, while mergers are on average the largest. All types of issues increase in frequency during our sample period. Strong investor protection laws are associated with more frequent public offerings and mergers, and fewer rights offerings. The relation between investor protection and private placements is positive, but less robust as compared to other types of issues. We relate these findings to several theories of issuance choice. Stock market development has no effect on seasoned issuance frequency or type. The findings are robust to excluding U.S. firms.

We study how seasoned shares are issued in 41 countries during the period 1990-2008. Our paper consists of two parts. In the first part we describe the frequencies and magnitudes of stock-financed mergers, rights offerings, private placements, and public offerings, which we define as all seasoned share issues that do not fall into one of the other three categories.<sup>1</sup> This section of the paper complements the analyses in Fama and French (2005), who describe how shares are issued in the U.S. We extend their analysis by studying share issues in 40 additional countries, and by creating a new share issues database. Like Fama and French and many other studies, we use share issues data from SDC Thomson. In addition, we conduct Factiva searches for news of share issues. Our Factiva searches increase the SDC sample by 43%.

In the second part of the paper we investigate how investor protection affects seasoned issues. La Porta, Lopez-de-Silanes, Shleifer, and Vishny (LLSV) (1997, 1998, and 2002) and La Porta, Lopez-de-Silanes, and Shleifer (LLS) (2006) posit that investor protection laws promote external finance by limiting expropriation and other activities that harm minority investors. These papers link investor protection to country-level measures of financial development, such as the size and liquidity of the stock market. In contrast, we study how investor protection and financial development affect the method and likelihood of seasoned issuance among existing public firms.<sup>2</sup>

We find that public offerings are the most common type of issue, followed by private placements, stock-financed mergers, and finally rights offerings. Excluding U.S. firms does not change the ordering. These findings are surprising, because there is a belief that outside of the

<sup>&</sup>lt;sup>1</sup> Firms can also issue shares indirectly through warrant exercises, stock purchase plans, and the conversions of debt and preferred stock. We do not have international data for these types of issues, so they are not included in our analyses.

<sup>&</sup>lt;sup>2</sup> Rajan and Zingales (2003) explain that countries with large stock markets need not be the same countries in which public firms actively issue shares. Rajan and Zingales (2003) point out that the presence of a few large companies that have appreciated in market value can give the impression of an important equity market even when the amount of equity issued in the market is very small.

U.S. seasoned shares are issued primarily through rights offerings or private placements.<sup>3</sup> All types of issues increase throughout our 19-year sample period; the percentage of firms making at least one share issue increases from 1.19% to 10.40%, while the average issue size increases from \$91.49M to \$231.92M. These findings are also surprising, because there is a belief that rights offerings have declined internationally, yet we find that the percentage of firms issuing by rights increases during our sample period.<sup>4</sup> Frank and Goyal (2003), Fama and French (2005), and McLean (2010) report increases in total share issues of U.S. firms during this period.

SDC issues data are widely used in the finance literature, so we report on the completeness of SDC coverage relative to our manual Factiva searches. Our Factiva searches increase the SDC samples by 51.1%, 40.1%, 44.2%, and 16.7%, for public offerings, mergers, private placements, and rights offerings respectively. SDC coverage gets worse over time for public offerings and mergers. Our Factiva searches increase the SDC samples by 35% and 29% for public offerings and mergers in 1990, and by 53.4% and 149.5% in 2008. SDC coverage tends to be more complete for wealthier countries and for countries with more developed stock markets, although SDC still misses a large number of issues in these markets.

Investor protection plays an important role in whether and how firms decide to issue shares. Firms in countries with strong investor protection laws issue more shares than firms in countries with weak laws, although this effect differs greatly across the issue types. Public offerings and mergers increase strongly with investor protection; private placements also increase, although the effect is not as robust, and rights offerings decline. Conditional on issuing, firms in weak investor protection countries are more likely to issue shares by rights offerings.

<sup>&</sup>lt;sup>3</sup> As an example, Cronqvist and Nilsson (2005) mention that in stock markets outside the U.S. seasoned equity offerings are conducted primarily as rights offerings and private placements.

<sup>&</sup>lt;sup>4</sup> The international decline in rights offerings is mentioned in Eckbo and Masulis (1995), Armitage (1998), and Eckbo, Masulis, and Norli (2007).

We include real per capita GDP in all of our regressions and find that it generally has the same sign and significance as the investor protection variables, showing that investor protection and economic development have independent effects on seasoned issuance. Financial development measures generally have no effect on seasoned issuance frequency or type. This shows that large stock markets are not the most active with respect to seasoned share issues. This finding is consistent with Demirguc-Kunt and Maksimovic (1998), who show that the percentage of firms in a country that rely on external finance to grow increases with investor protection, but not stock market development.

Our public offerings and stock-financed mergers findings suggest that legal protections are necessary for an active market in public offerings, which is consistent with the arguments in LLSV (1997, 1998, and 2002) and LLS (2006). The cross-listings literature also suggests that investor protection is a prerequisite for selling shares to minority shareholders. Papers by Coffee (1999, 2002), Stulz (1999), Reese and Weisbach (2002), Doidge, Karolyi, and Stulz (2004), and Doidge, Karolyi, Lins, Miller, and Stulz, (2009) posit that firms in weak investor protection countries cross-list in strong investor protection countries so that management can be bonded by the laws of the strong investor protection country. Such bonding is thought to increase the firm's market value and its access to equity financing. Our stock-financed merger results are consistent with Rossi and Volpin (2004), who show that mergers are more common in countries with strong investor protection.

Our finding that private placements are less affected by investor protection is consistent with previous studies, which suggest that the effects of the law on private placements are unclear. On the one hand, the law protects private placement investors from controlling shareholders, so we might expect private placements to increase with the investor protection. Alternatively, Wruck (1989) contends that private placements are purchased by active investors who monitor the firm in an effort to ensure that it uses its resources efficiently. Hertzel and Smith (1993) posit that private placements are issued to certify the firm's value. Barclay, Holderness, and Sheehan (2007) claim that entrenched managers place equity with friendly investors, who will not interfere with management's control. Monitoring, certification, and the benefits of control are all more valuable in low investor protection environments, so these studies therefore suggest that private placements should decrease with investor protection.

Our finding that rights offerings are more common in countries with weak investor protection is also consistent with previous studies. Zingales (1995), Bebchuk (1999), and Foley and Greenwood (2009) point out that investor protection reduces the benefits of control. Hence, controlling shareholders ought to have a stronger preference for rights in weak investor protection environments, in which control is more valuable.<sup>5</sup> Cronqvist and Nilsson (2005) use this framework to explain why family owned firms in Sweden prefer rights offerings to private placements. Eckbo and Masulis (1992) contend that owners prefer public issues over rights due to diversification benefits. Investor protection is necessary for domestic diversification to be beneficial, so for this reason too a controlling shareholder has fewer incentives to sell large parts of the firm in weak investor protection countries. Eckbo and Masulis (1992) further show that if existing shareholders plan to retain most of their new shares, then rights offerings are the lowest cost method of issue.<sup>6</sup> If the retention of rights declines with investor protection countries.

<sup>&</sup>lt;sup>5</sup> Ownership concentration is greater in countries with weaker legal protections, so in these countries rights offerings consist largely of share sales to insiders. See La Porta, Lopez-de-Silanes, and Shleifer (1999) and Foley and Greenwood (2009).

<sup>&</sup>lt;sup>6</sup> Eckbo and Masulis (1992) contend that if the market expects existing shareholders to sell the rights or new shares acquired though the rights, then the market infers that the shares were overpriced, and you then end up

Our study complements international share issues studies by Henderson, Jegadeesh, and Weisbach (2006) and Kim and Weisbach (2008). Henderson, Jegadeesh, and Weisbach (2006) study the use of equity versus debt, common versus preferred stock, and offerings on foreign exchanges versus depository receipts. Kim and Weisbach (2008) focus on the motivations for share issues by studying the uses of public offerings proceeds. In contrast, we study different issue mechanisms, and the effects that investor protection and financial development have on overall issues and the choice of issue mechanism.

The remainder of this paper is organized as follows. Section 1 discusses the data and sample used in this paper; Section 2 provides detailed descriptions of how shares are issued around the world; Section 3 analyzes the role of law and legal environment in the choice of share issuance method; Section 4 concludes the paper.

#### **1.** Data Sources and Sample Description

We begin our sample with all Worldscope firms during the period 1990 to 2008. We begin our sample with Worldscope firms because a good deal of our analyses requires firm-level accounting data. Like Kim and Weisbach (2008), we exclude utility firms with 3-digit SIC codes from 491 to 494, financial firms with 1-digit SIC code 6, and telecommunication firms with 2-digit SIC code 48. We focus a good deal of our analyses on how investor protection affects issuance activities. Our sample countries are therefore limited to countries that have investor protection measures from LLS (2006). We add an additional country-level filter that requires at least 20 share issues in total over the sample period. A focus of the paper is how investor

with the type of adverse selection problems and underpricing effects that Myers and Majluf (1984) model with public offerings. Consistent with this view, Bohren, Eckbo, and Michaelsen (1997) provide evidence with Norwegian firms that the use of rights decreases with the propensity for shareholders to sell their rights.

protection laws affect issuance activity, so for this reason we only include domestic share issues in our sample. As mentioned in the Introduction, several papers study the importance of investor protection with respect to cross-listings.

We use two different data sources for share issues. Our first source is SDC Thompson Financial. The SDC Global New Issues Database provides data for non-merger share issues, while the SDC Mergers & Acquisitions Database provides data for stock-financed mergers. SDC has very limited coverage for non-U.S. share issues prior to 1990, so our sample period is from January 1990 through December 2008. Thomson representatives told us via phone conversations that they felt their coverage in both databases is robust from 1990 onwards. Kim and Weisbach (2008) also begin their global public offerings sample in 1990.

We begin our non-merger share issues sample with all domestic seasoned (non-IPO) share offerings in the SDC Global New Issues Database. The SDC Global New Issues data provides us with data on how the issue was made – through either public offerings, or private placements, or rights offerings. We eliminate observations that have missing information on either the total number of shares sold, or the breakdown of these shares into primary and secondary shares. We exclude secondary share issues, since secondary issues involve the sale of existing shares that are owned by insiders, and do not increase the number of shares outstanding as primary issues do. We follow Kim and Weisbach (2008), and exclude issues that have values that are different from the product of the offer price and the number of shares sold by more than \$5 million.

We begin our merger-issues sample with all domestic mergers in the SDC Mergers & Acquisition Database. We include all M&A transactions that were completed between the years 1990 to 2008. We measure the value of a stock-financed merger as the portion of the total

transaction value that is paid with the acquiring firm's stock. We exclude observations if either the M&A deal value or the percentage of stock payment for the deal is missing.

Our second source of share issues data is Factiva news searches. We conduct English language news searches for every firm in our Worldscope sample during the years 1990 through 2008. We conduct our search by first specifying the firm and country along with one of the following words: "acquisition", "merger", "offering", or "placement". If the search returns an excessive number of news stories then we do more refined searches using the terms: "merger and acquisition", or "rights offering", or "private placement", or "public offering". To be included in our sample the news story had to include the issue amount and date.

It is likely that our news searches miss some issues, so our searches provide a lower bound on the number of issues that SDC omits. Given that our news searches are conducted in English, it is likely that we miss more issues in countries where English is not the primary language. In such countries we expect to mainly miss smaller issues that are less likely to be mentioned in English by international news sources. Our final sample consists of 59,029 share issues.

World Scope reports accounting information that can be used to estimate total share issues. McLean, Pontiff, and Watanabe (2009) use Data Stream data to estimate real changes in shares outstanding, which also captures total share issues. We chose not to use share issues measures from these sources because the objectives in this paper include studying how differences in investor protection relate to different issue types. Share issues measures from World Scope and Data Stream reflect aggregate issues, and therefore do not show how the shares were issued (e.g. public offering or private placement). In addition to the four types of issues that we study, these aggregate issuance measures also include issues from warrant exercises, stock purchase plans, and the conversions of debt and preferred stock. We have regressions in the paper in which total share issuance (the sum of public offerings, mergers, rights offerings, and private placements) is regressed on different measures of investor protection and development. For these tests we obtain similar results with a total share issuance measure from World Scope.

#### 2. A Description of how Shares are Issued Around the World

# 2.1. Frequency and Average Amount of each Type of Issue

Table 1 describes global share issues in each of the nineteen years in our sample. Panel A reports the number of issues, Panel B reports the percentage of firms that issue, while Panel C reports the average issue size. The findings in Panel A show that public offerings are the most common type of share issue. In total, our sample has 21,263 public offerings, 18,145 private placements, 14,961 stock-financed mergers, and 4,660 rights offerings. This ordering does not change if U.S. firms are excluded from the sample. As we mention in the Introduction, these findings are surprising, because there is a belief that outside of the U.S. shares are typically issued by rights offerings or private placements, yet here we find that globally public offerings are the most frequent type of issue.

Panel B shows the percentage of firms issuing shares in each year. Overall, 6.88% percentage of the firm-year observations in our sample issue shares via at least one of the four issue mechanisms. 2.93% of the firm-year observations issue via public offerings followed by mergers (2.09%), private placements (2.01%), and rights offerings (0.65%). Panel A shows that there are more private placement issues than merger issues, yet here we see that the percentage of firm-year observations issuing via mergers is greater. This difference arises because in our sample it is more common for a firm to make several private placements in the same year than it

is for a firm to issue shares in several different mergers in the same year.

Panel A shows that there is a pronounced increase in the number of issues, as our sample has 261 total issues in 1990 and 5,585 in 2008. Panel B shows that the percentage of firms issuing shares also increases during the sample period. In 1990 only 1.19% of the firms in our sample issued shares, while in 2008 10.4% of the firms in our sample issued shares. In both Panels A and B the increases are caused by secular trends, which are observed with all four issue types. It is unlikely that these results are caused by better data coverage in the later years of our sample, because our sample consists of both SDC data and manual news searches, and we use the same search methods for each year. Moreover, the increases documented here are consistent with increases in U.S. share issues documented in Frank and Goyal (2003), Fama and French (2005), and McLean (2010). The patterns of increasing issues observed in Panels A and B persist if U.S. firms are excluded. The results here reveal a worldwide increase in seasoned share issues.

Panels A and B both show that rights offerings increase steadily throughout the sample period. As we mention in the Introduction, this increase in rights offerings is surprising, because previous studies have mentioned that the use of rights offerings has declined during our sample period.

Panel C reports the average size of each type of issue in real (2005) U.S. dollars. On average, mergers are the largest issue (\$340.83M), followed by public offerings (\$118.93M), rights (\$79.33M), and private placements (\$30.57M). The size of the average merger and average public offering increases over the sample period, while the size of the average private placement and average rights offering do not trend in either direction over the sample period. The average merger increases 200%, from \$114.99M to \$345.58M, while the average public offering increases 120%, from \$65.59M to \$144.56M. Both of these increases appear to be the

result of secular trends, with both variables showing steady increases throughout the sample period.

### 2.2. The Completeness of SDC Coverage

Table 2 reports the increase in our SDC issues sample due to the inclusion of the issues that we found in our Factiva searches. The numbers reported in Table 2 are the number of issues that we found in Factiva scaled by the number of issues reported by SDC. Our Factiva issues do not include issues that are also reported in SDC, so the numbers in Table 2 reflect the percentage increase in our SDC sample due to the inclusion of the issues obtained from Factiva searches. A higher percentage reflects less complete SDC coverage.

In a few countries, for a type of issue SDC reports zero, but our Factiva searches found one or more issues. In these instances, we report "N/A" for not applicable, as we cannot compute a ratio in these instances. Each country has more than one SDC issue in total, so we are able to report a ratio for total issues for each country.

The results reported in Table 2 show that overall our Factiva searches increase our SDC sample by 42.9%. The Factiva searches increase the SDC samples by 51.1%, 40.1%, 44.2%, and 16.7%, for public offerings, mergers, private placements, and rights respectively. It is likely that our Factiva searches miss some issues, so these findings probably overstate the completeness of SDC coverage.

Surprisingly, the results in Table 2 show that SDC coverage gets worse over the sample period for some types of issues. Our Factiva searches increase the SDC merger sample by 29% in 1990 and by 149.5% in 2008. This increase appears to be a secular trend, with SDC coverage getting steadily worse over the sample period. SDC coverage for public offerings also gets worse

over the sample period, although the effect is not as great as with mergers. During the first five years of the sample, the Factiva searches increase the SDC public offerings sample by about 32.6%, whereas the increase averaged 70.3% during the most recent five years.

In 1990 SDC coverage was very poor for private placements and rights offerings; our Factiva searches increase these samples by 2500% and 900% respectively in this year. SDC coverage then improves over the next few years for both types of issues. From the late 1990s through the final year of the sample there is no discernable trend in coverage for either issue. In the final year of the sample, the Factiva searches increase our SDC rights offerings sample by 16.8%, and our SDC private placements sample by 37.7%.

## 2.3. Issue Frequency Summarized by Country

Table 3 describes issue activity within each of the 41 countries in our sample. Panel A lists the frequency of each issue, while Panel B lists the percentage of firm-year observations that issue shares. Panel A shows that the U.S. is the largest contributor to our sample, accounting for 33.2% of the total. The next largest contributor is Australia, which accounts for 21.6% of the total. Australia's issues consist mainly of 9,213 private placements, which account for almost half of the private placements in our sample. Kim and Weisbach (2008) also note that more than half of all private placements come from Australia.<sup>7</sup> The next two largest contributors to our sample are Canada and Japan, accounting for 8.3% and 7% of the total issues.

Panel B displays the percentage of firm-year observations that issue shares for each of the countries in our sample. The countries with the highest percentage of issuers are Australia (24.44%), Canada (15.39%), and Singapore (10.34%). As we mention above, in Australia the

<sup>&</sup>lt;sup>7</sup> Our analyses are robust to the exclusion of Australian and U.S. issues.

high issue frequency in each of these countries is largely due to private placements. In Canada and Singapore public offerings are the most common share issue. The countries with the lowest percentage of issues are Pakistan (1.01%), Turkey (1.16%), and Greece (1.33%). As comparisons, the percentage of issuers is 7.58% in the U.S. and 6.88% in the entire sample.

### 2.4. SDC Completeness Summarized by Country

Table 4 reports the Factiva enhancement for each issue type within each country. As in Table 2, the numbers in Table 4 reflect the percentage increases in our SDC samples due to our Factiva searches. Table 4 shows that SDC coverage is weakest in Pakistan, Israel, Mexico, and India, where our Factiva searches increased the SDC issue samples by 2200%, 544%, 418%, and 384% respectively. The total number of issues in Pakistan is only 23, so the increase due to our Factiva searches does not reflect a large number of issues in the case of Pakistan. The total numbers of issues in Israel, Mexico, and India are 219, 311, and 1,917, so our Factiva searches uncover a large number of issues in these countries.

SDC coverage is strongest in Australia (8% increase), Venezuela (16% increase), Chile (18% increase), and Japan (22.7% increase). Yet even in these countries the percentages can reflect a large number of issues missing from the SDC sample. In Japan, the total number of issues is 4,129, so 764 Japanese issues were uncovered by our Factiva searches. In the U.S., our Factiva searches increase the sample by 52.8%, showing that SDC missed more than half of the seasoned issues in the word's largest stock market. In the U.S., SDC coverage is the best for mergers, as the merger sample increases by 33.9%, and worse for private placements, as that sample increases by 98.1%.

In unreported tests we regress our SDC-Factiva country-statistics on per capita GDP and

measures of stock market development (stock market capitalization as a percentage of GDP and dollar trading volume scaled by total stock market capitalization). We find that our Factiva searches increase the SDC issues samples more in low GDP countries, and more in countries with less developed stock markets, showing that SDC coverage is typically less complete in these countries.

# 3. Cross-Country Differences in Seasoned Issues and Choice of Issue Method

In this Section we study how cross-country differences in investor protection and development affect the seasoned issues. In Section 3.1 we discuss our empirical predictions for each issue type. Section 3.2 describes our investor protection and development measures. Section 3.3 discusses results from country-level regressions, while Section 3.4. discusses the findings from firm-level regressions.

# 3.1. The Law and Cross-Country Differences in Share issues: Empirical Predictions

As discussed in the Introduction, the idea that investor protection ought to lower the cost of issuing shares is developed in LLSV (1997, 1998, and 2002) and LLS (2006). The argument is that without proper legal protections managers and controlling shareholders will expropriate the firm's assets at the expense of minority shareholders. Potential shareholders know this, and either avoid buying shares, or are only willing to buy shares at discounted prices that reflect the potential loss from expropriation. Consistent with this framework, LLSV and LLS show that financial development and market values are increasing with investor protection.

This framework also predicts that investor protection promotes seasoned issues, although we posit that the relation ought to be different for different types of issues. As Rajan and Zingales (2003) explain, measuring seasoned issues is different from measuring financial development. Financial development typically refers to the size of the stock market or credit market relative to GDP. In this paper we study seasoned issues by existing public firms. If it is typically less costly for public firms to issue seasoned shares in larger stock markets, then we ought to see share issues increase with financial development. The findings in Demirguc-Kunt and Maksimovic (1998) suggest that this may not be the case, as they show that the percentage of firms in a country that rely on external finance to grow increases with investor protection, but not stock market size.

*Public Offerings*. In a public offering the new shareholders will most likely only have small levels of ownership. For such investors laws that protect against expropriation are important, and public offerings should therefore increase with investor protection. Moreover, Leuz, Nanda, and Wysocki (2003) show that investor protection laws encourage more accurate financial reporting, while Morck, Yeung, and Yu (2000) contend that investor protection encourages more arbitrage, which results in stock prices that more accurately reflect fundamental values. The findings in both of these papers suggest that the adverse selection risks described in Myers and Majluf (1984) are lower in countries with stronger investor protection. Hence, small, relatively uninformed investors face less risk when buying seasoned shares in countries with strong investor protection, so for this reason too public offerings should increase with investor protection.

Zingales (1995), Bebhuck (1999), and Foley and Greenwood (2009) point out that investor protection reduces the benefits of control. Hence, the incentives to maintain a controlling interest are lower in countries with strong investor protection laws. Owners should therefore be more willing to sell shares to the public in countries with stronger investor protection, so for this reason too we expect public offerings to increase with investor protection. Consistent with this reasoning, Foley and Greenwood (2009) show that post-IPO ownership concentration reduces faster in countries with strong investor protection.

*Stock-Financed Mergers*. Investor protection laws should also promote stock-financed mergers and acquisitions. The arguments here are similar to those with public offerings. In a stock-financed merger, a portion of the target firm is sold to the acquirer in exchange for the acquirer's shares. If laws reduce expropriation, improve financial reporting, and encourage arbitrage, then the typical acquirer should have more valuable shares from the perspective of the target's shareholders. Hence, stock-financed mergers ought to be a more financially viable alternative from the acquirer's perspective in countries with strong investor protection. Investor protection also reduces the benefits of control, making owners more willing to issue shares to outsiders.

Consistent with these views, Rossi and Volpin (2004) show that a firm is more likely to be acquired if it is located in a county with strong investor protection, and that all-cash mergers are less common if the acquirer is from a country with strong investor protection. Rossi and Volpin's unit of observation is a transaction in which ex-ante the acquirer owned less than 50% of the firm, and ex-post the acquirer owned more than 50%. Our interest is measuring how firms issue shares, so our unit of observation is shares issued in any merger. This difference in measurement leads to our having a much different sample from Rossi and Volpin's. In total, our merger-share issuance sample (1990-2008) consists of 14,961 merger-related share issues, while Rossi and Volpin's sample (1990-2002) consists of 4,007 change-of-ownership merger transactions.

15

*Private Placements.* As we mention in the Introduction, the expected effects of the law on private placements are ambiguous. The law protects private placements investors from controlling shareholders, so we might expect private placements to increase with the law. However, private placement investors tend to be larger and more sophisticated as compared to the typical minority investors, so for this type of investor legal protections may be less important. Moreover, there are at least three theories regarding why firms chose to issue via private placements that suggest that private placements should decrease with investor protection. Wruck (1989) contends that private placements are purchased by active investors who are willing and able to monitor the firm and ensure that it uses its resources efficiently. Hertzel and Smith (1993) posit that private placements are used to certify the firm's value. Barclay, Holderness, and Sheehan (2007) claim that entrenched managers place shares with friendly investors who will not affect management's control. Monitoring, certification, and the benefits of control are all more valuable in low investor protection environments, so these three theories predict that private placements should decrease with the law.

Jang, Kim, and Koo (2010) show that private placements are the most common type of issue in Korea, where private placements are typically used to reshape ownership structure in business groups. If business groups are more common in countries with weaker investor protection, then Jang et al.'s findings also suggest that private placements ought to be more common in countries with fewer investor protections.

*Rights Offerings*. Rights offerings give existing shareholders the right to buy additional shares from the firm. In contrast to other share issues, we expect the use of rights to decline with investor protection. We outline this reasoning in the Introduction. In weak investor protection environments rights offerings may be the choice of issue by default, because the public is simply

unwilling to buy shares due to the low levels of legal protections. By the same logic, weak investor protection encourages entrenched owners to issue via rights offerings, so that their ownership and the benefits that it provides them are not diluted (Zingales (1995), Bebchuk (1999), and Foley and Greenwood (2009)). Eckbo and Masulis (1992) contend that owners typically prefer public issues over rights, due to diversification benefits. However, if legal protections are low, then domestic diversification is less beneficial, making rights more attractive relative to other types of share issues. Eckbo and Masulis (1992) and Eckbo (2004) also show analytically that if existing shareholders plan to retain most of their new shares, then rights offerings are the lowest cost method of issuing shares. If the retention of rights declines with investor protection, then rights offerings should decline with investor protection according to the Eckbo and Masulis (1992) framework.

## 3.2.2. Measures of Investor Protection and Development

To test for a link between investor protection and seasoned share issues we use five different measures of investor protection. Our objective is not to run "horse races" between the different measures of investor protection, but rather test for a robust relation between share issues and investor protection using several different investor protection measures, which have been shown to be important for finance in previous studies. The investor protection variables are summarized in Table 5.

*Common.* Johnson, La Porta, Lopez-de-Silanes, and Shliefer (2000) explain that civil law courts are poorly suited for disciplining controlling shareholders accused of self-dealing transactions such as tunnelling, whereas the common law system, which allows greater judicial discretion in determining the appropriateness of self-dealing transactions, yields courts that

afford better protection to minority shareholders. LLSV (1998), LLS (2006), and Djankov et al (2008) show that common law countries tend to have stronger investor protection laws and greater enforcement of these laws than civil law countries. We therefore create a dummy variable *Common* that is equal to 1 if a country is of common law origin and zero if the country is of civil law origin.

*Disclosure, Liability,* and *Protect.* LLS (2006) show that disclosure requirements and liability standards are more strongly associated with financial development than other legal factors. We therefore use their disclosure and liability indices as measures of investor protection. *Disclosure* is the arithmetic mean of six sub-indices. One of these sub-indices indicates whether or not new issues need to be accompanied by a prospectus. The five other sub-indices measure disclosure requirements within the prospectus regarding directors' and officers' compensations, controlling shareholders, insider ownership, irregular contracts, and any transactions between the issuer and its officers and directors. *Liability* is the arithmetic mean of three sub-indices. The sub-indices measure the ease with which an investor can pursue an issuer and its directors, the distributors, and the accountants in civil court if the investor suffers losses due to misleading statements in a prospectus. LLS (2006) also develop a composite investor protection index, which we refer to as *Protect. Protect* is the first principal component of *Liability, Disclosure*, and an index of anti-director rights.

*Efficiency of the Judiciary* measures the "efficiency and integrity of the legal environment as it affects business, particularly foreign firms." *Efficiency of the Judiciary* ranges in value from 0 to 10, with 10 meaning the best legal environment. The measure is created by International Country Risk, an agency that rates country risk. LLS (2006) show that higher

values of *Efficiency of the Judiciary* are associated with larger stock markets, more publicly traded firms, and greater access to equity for small and medium-sized firms.

*Stock Market* and *Turnover*. Share issues could also be less costly in countries with more developed stock markets. We therefore test whether proxies for equity market development are related to share issues. Our equity market development measures were obtained from Andrei Shleifer's Web site. *Stock Market* is the aggregate market capitalization of all publicly traded firms scaled by total GDP. *Turnover* is the total dollar value of stocks traded, scaled by total stock market capitalization, for the period 1996-2000.

*GDP*. Wealthier countries should have more citizens with the means to invest, so for this reason we expect that higher real per capita gross domestic product (*GDP*) ought to be associated with more frequent share issues. *GDP* is correlated with investor protection and financial development, so we use *GDP* both by itself and as a control variable in our regressions. We use the average of the yearly values of real (2005 USD) per capita *GDP* for each of the countries in our sample, although we obtain similar results if *GDP* is measured only at the beginning of our sample period.

## 3.3. Country-Level Regression Tests

Table 6 reports the results from our country-level regressions. The dependent variable is the percentage of firm-year observations in each country that made at least one issue. In unreported tests we estimated regressions in which the dependent variable is the firm's issue amount scaled by lagged assets, averaged at the country-level, and obtained similar findings. Tstatistics are computed with White (1980) heteroscedasticity-consistent standard errors.

In the first regression GDP is the only independent variable, and its coefficient is positive

and statistically significant, showing that firms located in wealthier countries are more likely to issue seasoned shares. To get an idea of the economic significance, *GDP* has a standard deviation of 0.859. In the first regressions the GDP coefficient is 0.013 (*t*-statistic = 1.94), so a one standard deviation increase in GPD yields a 1.1% increase (e.g. 2% to 3.1%) in the percentage of firms that issue shares.

The next five regressions include each of the investor protection measures along with *GDP*. The coefficients for all five of the investor protection measures are positive and statistically significant, suggesting a causal relation between investor protection and share issues. The effects are economically significant as well. As an example, the *Common* coefficient is 0.044 (*t*-statistic = 2.98), so the percentage of firms issuing shares in common law countries is on average 4.4% greater that in civil law countries. *GDP* remains positive and significant in most of these regressions, suggesting that a country's wealth and the level of its investor protection have distinct effects on the cost of issuing shares.

The final two regressions include the stock market development measures along with *GDP*. The *Stock Market* variable is not statistically significant, while the *Turnover* variable is negative and significant. These results show that larger stock and even more liquid stock markets are not necessarily more active in terms of seasoned issuing shares.

### 3.4. Firm-Level Regressions

In this Section of the paper we discuss results from firm-level Probit regressions. An advantage of firm-level regressions relative to country-level regressions is that we control for firm-level characteristics that are likely to affect whether and how firms issue shares. The dependent variable in these regressions is equal to 1 if the firm issued shares during the year, and zero otherwise. We make this variable for total issues and each of the four different issue types separately. The regressions include both industry (1-digit SIC code) and year fixed-effects. Per the recommendations in Petersen (2009) the standard errors are clustered on the country.

In unreported tests we estimate Tobit regressions and have similar findings. The effects that we describe below therefore apply to both the probability of issuing, and issue amount conditional on issuing. The Tobit results are available from the authors upon request.

Our choice of firm-level control variables follows Rajan and Zingales (1995) and Baker and Wurgler (2002), who regress changes in leverage on firm characteristics. We believe that these control variables might also matter for issue type. Moreover, in some cases these firm-level variables are correlated with the investor protection measures, so controlling for these firmcharacteristics may be important with respect to measuring the influence that investor protection has on share issues.

Our firm-level control variables are as follows. Firm size (*Assets*) is measured as the log of total assets. Smaller firms tend to grow faster, and therefore should need more capital. Market-to-book ratio (*MB*) is estimated as the market value of equity, minus the book value equity, plus the book value of assets divided by the book value of assets. *MB* reflects the firm's growth opportunities. Market value of equity is measured as year-end stock price multiplied by year-end shares outstanding, while book value of equity comes form the firm's financial statements. All else equal firms with larger growth opportunities ought to issue more shares. Loughran and Ritter (1995) and Baker and Wurgler (2002) contend that *MB* may also reflect mispricing. Hence, *MB* may also predict share issues because firms issue shares in response to mispricing. *Leverage* is total debt scaled by total assets. *PPE* is property, plant and equipment total scaled by assets. *PPE* measures the portion of the firm's assets that cannot be easily converted to cash. *CF* 

is net cash flows from operations scaled by assets measured at the beginning of the year. All else equal, firms with lower cash flow ought to need more capital, and should therefore issue more shares.

# 3.4.1. Aggregate Issues

Our firm-level regression results are reported in Tables 7-11. In Table 7 the dependent variable is equal to 1 if the firm made at least one share issue of any type and zero otherwise. The results in Table 7 show that in firm-level regressions *GDP* does not consistently predict more frequent share issues; however each of the five investor protection coefficients is positive and statistically significant, showing that greater levels of investor protection are associated with more frequent share issues. These effects are economically significant as well. As an example, the *Disclosure* coefficient is 0.074 (*t*-statistic = 2.71). The coefficient reflects a marginal probability, and *Disclosure* has a standard deviation of 0.216. Hence, a one standard deviation increase in *Disclosure* yields an increase of 1.6% in the likelihood that a firm will issue at least one type of share during the year. Panel B of Table 1 shows that 6.88% of the firms in our sample make at least one issue per year, so investor protection has a relatively large effect.

As with the country level tests, neither of the financial development measures are significant, showing that countries with large financial markets do not necessarily have more seasoned offerings. As we mention in the Introduction, these findings are consistent with Demirguc-Kunt and Maksimovic (1998), who show that the percentage of firms in a country that use external finance to grow increases with investor protection, but not financial development. Levine and Zervos (1998) document similar effects at the macroeconomic-level, as they find no link between stock market size and economic growth.

With respect to the firm-level controls, the coefficients reported in Table 7 show that the relations between share issues and market-to-book, leverage, and *PPE* are all positive. This means that firms with stronger growth opportunities, high leverage, and fewer assets that can be easily converted to cash issue more shares. The relation between share issues and cash flow is negative, showing that less profitable firms issue more shares. McLean (2010) reports similar relations in a sample of U.S. firms.

# 3.4.2. Public Offerings

Table 8 reports the results from the public offerings regressions. The results show that GDP is associated greater public offerings. The GDP coefficient is positive and significant in the regression without any of the other measures, and in six out of the seven regressions that include either a protection or development measure. Hence, firms issue more shares via public offerings in wealthier countries, and this effect is observed after controlling for the effects of investor protection and financial development. All five of the investor protection coefficients are positive and statistically significant, showing that firms issue more shares via public offerings in countries with strong investor protection laws. As for economic significance, the *Common* coefficient is 0.008 (t-statistic = 2.00), showing that a firm in a common law country is 0.8% more likely to issue via a public offering than is a firm in a civil law country. Panel B of Table 1 shows that in our sample 2.93% of the firms issue by public offerings in a given year, so an increase of 0.8% in the likelihood of issuing is a sizeable effect.

The findings here are broadly consistent with the notions that laws reduce information asymmetries, expropriation, and adverse selection problems, thereby making public issues more feasible. The results are therefore consistent with the LLSV framework, which posits that investor protection is necessary to facilitate the sale of securities to minority investors. Neither of the stock market development measures is associated with greater public offerings, as both of the coefficients are insignificant. Hence, large and even liquid stock markets are not associated with more public offerings.

### 3.4.3. Stock-Financed Mergers

The results in Table 9 show that firms in countries with higher *GDP* and higher levels of investor protection are more likely to issue shares in mergers. As with public offerings, the investor protection results are very robust, as all five of the investor protection coefficients are positive and statistically significant. As an example, the *Liability* coefficient is 0.027 (*t*-statistic = 6.67), showing that a one standard deviation increase in *Liability* yields a 0.70% increase in the likelihood that a firm issues shares in a merger. Panel B of Table 1 shows the percentage of firm-year observations with merger share issues is 2.09%, so investor protection has a large effect on the use of shares in mergers. The findings are consistent with the notions that laws reduce information asymmetries, expropriation, and adverse selection problems, thereby making equity a more feasible currency in mergers and acquisitions.

As with public offerings, neither the *Stock Market* nor the *Turnover* coefficient is significant, showing that stock market size and liquidity are not associated with the amount of shares that are issued in mergers.

### 3.4.4. Private Placements

Table 10 reports the results for private placements. *GDP* is significant in only one of the regressions, so it does not have a robust effect on private placements. The *Common* and *Protect* 

coefficients are both positive and significant, however the three other investor protection indices are insignificant. The results suggest that investor protection has some impact on the frequency of private placements, but not to the same extent as with public offerings and stock-financed mergers. These findings are sensible, for as we discuss in the Introduction and in Section 3.1 of this paper, private placements are typically made to large, well-informed investors, who face fewer information asymmetries, so legal protections may be less important to these investors relative to smaller, individual investors. Moreover, papers by Wruck (1989), Hertzel and Smith (1993), and Barclay, Holderness, and Sheehan (2007) stress the roles of monitoring, certification, and the benefits of control respectively in motivations for private placements. Each of these effects is probably more valuable in low investor protection countries, so these papers suggest that private placements should decrease with the law.

With respect to the firm-level characteristics, the size coefficient is negative in all of the private placement regressions, showing that private placements are more likely to be done by smaller firms. This is in contrast to the results in Table 8, which shows that size has no effect on public offerings, and Table 9, which shows that merger-related issues are more likely to be done by larger firms. Hertzel and Smith (1993), Chemmanur and Fulghieri (1999), Cronqvist and Nilsson (2004), and Wu (2004) contend that firms with high levels of asymmetric information will be more likely to issue via private placements as opposed to public offerings, so our findings with respect to size are consistent with these papers.

# 3.4.5. Rights Offerings

Table 11 reports the results for rights offerings. The results are very different with rights as compared to the other types of issues. *GDP* is not associated with greater rights offerings.

With respect to the investor protection, only the disclosure and liability indices are significant, and both of the coefficients are negative, showing that rights are more common in countries with lower levels of investor protection. The findings therefore suggest that investor protection laws do not promote the use of rights, and may even discourage the use of rights.

As we discuss in the Introduction and in Section 3.1 of the paper, there are several reasons to suspect that rights offerings decline with investor protection. First, firms may choose rights in low investor protection countries because this is the only way that they can issue shares. Consistent with this view, the size coefficient is negative and significant in most of the rights offerings regressions. If information asymmetries are greater for smaller firms, then this effect again suggests that rights are used by firms that have difficulty raising capital from small investors. Second, entrenched owners may prefer to issue via rights offerings, so that their ownership and the benefits that it provides them are not diluted. Such control benefits are most likely higher in countries with low levels of investor protection. Third, Eckbo and Masulis (1992) contend that owners would prefer public issues over rights, due to diversification benefits. However, if legal protections are low, then the benefits of domestic diversification are also lower. Finally, Eckbo and Masulis (1992) show analytically that if existing shareholders plan to retain most of their new shares, then rights offerings are the lowest cost method of issuing shares. If the retention of rights declines with investor protection for the reasons listed above, then rights offerings should also decline with investor protection in the Eckbo and Masulis (1992) framework.

## 3.4.6. Rights vs. Other Types Equity Issues

A clear empirical prediction of the preceding discussion in the last Section is that

conditional on issuing, firms in low investor protection countries should be more likely to issue via rights relative to other share issuance methods. This hypothesis is tested in this part of the paper. The dependent variable in these regressions is the amount of capital raised in rights offerings scaled by the sum of capital raised in public offerings, rights offerings, and private placements (mergers are excluded, although we get similar results if mergers are included). The hypothesis tested is that conditional on issuing, a firm in a low investor protection country is more likely to choose a rights offering.

The results in Table 12 are consistent with the hypothesis that investor protection laws discourage rights and encourage other types of issues. Although the common law dummy variable is insignificant, the four other investor protection indices are negative and significant, showing that the propensity for issuers to choose rights offerings over other equity issuance methods declines with investor protection. The effects are economically significant as well. As an example, the *Disclosure* coefficient is -0.395 (*t*-statistic = -4.43), showing that a one standard deviation increase (0.220) in *Disclosure* yields an increase of 0.087 in the percentage of equity issued by rights.

The *GDP* coefficient is negative in all of the regressions and significant in five of the eight regressions, showing that firms in poorer countries do tend to favour rights over other equity sale methods. This may be because the potential investor base is much smaller in poor countries, and existing shareholders therefore represent a relatively large fraction of the investor base. As in the previous tables, neither of the financial development measures is significant, showing that the choice of rights versus other types of share issues is not a function of how large or active a stock market is.

# 4. Conclusion

This paper studies how shares are issued around the world. We conduct our study with a sample of firms drawn from 41 countries over a 19-year period. Our sample consists of share issues from SDC and a manual Factiva search. Our Factiva search increases the SDC issues sample by 43%. We provide a detailed description of the completeness of SDC coverage for each type of issue. We provide these descriptions over time and for each of the countries in our sample. We find that SDC coverage is typically more complete in developed countries and in countries with larger stock markets, although even in these countries SDC still misses a large number of issues.

We find that public offerings are the most common type of share issue, whereas stockfinanced mergers are on average the largest. Over time, seasoned share issues increase in frequency and in size; during our 19-year sample period the percentage of firms issuing shares increases ten-fold, while the average issue size more than doubles. Across countries, aggregate issues increase with investor protection and *GDP*, but not with measures of stock market development. The findings are consistent with the law and finance framework, which posits that investor protections are necessary in order for firms to sell large quantities of shares to minority investors. The findings are also consistent with Demirguc-Kunt and Maksimovic (1996), who show that the percentage of firms in a country that rely on external finance to grow increases with investor protection, but not stock market size.

The effects of investor protection are strongest on mergers and public offerings. Both types of issues typically involve share sales to a large number of small investors, so this finding is consistent with the notion that investor protection is a prerequisite for broad investor participation. The effect of the law on private placements is weaker, suggesting that monitoring, certification, and maintaining control may be motivations for issuing via private placements. In contrast to other issues, rights offerings are more common in countries with fewer investor protections. This is consistent with rights being preferable when the cost of selling shares to the public is high, the benefits of control are high, and the likelihood that the rights are sold to the public is low.

# References

- Armitage, Seth, 1998, Seasoned equity offers and rights issues: A review of the evidence, The European Journal of Finance, Vol (4), 29-59.
- Baker, Malcolm, and Jeffrey Wurgler, 2002, Market timing and capital structure, Journal of Finance 57, 1-32.
- Barclay, Michael J., Clifford G. Holderness, and Dennis P. Sheehan, 2007, Private placements and managerial entrenchment, Journal of Corporate Finance, 13, 461-484.
- Bebchuk, Lucian. 1999, A Rent Protection Theory of Corporate Ownership and Control, Working Paper 7203, National Bureau of Economic Research.
- Bohren, Oyvind, Eckbo, Espen, and Dag Michalsen, 1997, Why underwrite rights offerings? Some new evidence, Journal of Financial Economics 46, 223-261.
- Chemmanur, Thomas J. and Paolo Fulghieri, 1999, A Theory of the Going-Public Decision, Review of Financial Studies, 12, 249-279.
- Coffee, John C., 1999, The future as history: The prospects for global convergence in corporate governance and its implications, Northwestern Law Review 93, 641–707.
- Coffee, John C., 2002, Racing towards the top? The impact of cross-listings and stock market competition on international corporate governance, Columbia Law Review 102, 1757–1831.
- Cronqvist, Henrik, and Mattias Nilsson, 2005, The choice between rights offerings and private equity placements, Journal of Financial Economics 78, 375-407.
- Demirguc-Kunt, A., and Maksimovic, V., 1998, "Law, Finance and Firm Growth." Journal of Finance 53, 2107-2137.
- Doidge, Craig, Andrew Karolyi, and Rene Stulz, 2004, Private Benefits of Control, Ownership, and the Cross-listing Decision, Journal of Finance 64, 425-466.
- Doidge, Craig, Andrew Karolyi, Karl Lins, Darius Miller and Rene Stulz, 2009, Why Are Foreign Firms Listed in the U.S. Worth More? Journal of Financial Economics 71, 205-238.
- Eckbo, B. Espen, 2004, Equity Issues and the Disappearing Rights Offer Phenomenon, Journal of Applied Corporate Finance 20 (4), 72-85.
- Eckbo, B. Espen, Masulis, Ronald W., and Øyvind Norli, 2007, Security Offerings, Handbook of Corporate Finance: Empirical Corporate Finance, Volume 1, (Elsevier/North-Holland Handbook of Finance Series), Ch. 6, 233-373.

- Eckbo, B. Espen and Ronald W. Masulis., 1995, Seasoned Equity Offerings: A Survey, 1995, inR. Jarrow, V. Maksimovic and W. Ziemba (eds.) Finance (North-Holland, Series of Handbooks in Operations Research and Management Science), Ch. 31, 1017-1072.
- Eckbo, B. Espen and Ronald W. Masulis, 1992, Adverse Selection and the Rights Offer Paradox, Journal of Financial Economics 32, 293-332.
- Fama, Eugene and Kenneth R. French, 2005, Financing Decisions: Who Issues Stock? Journal of Financial Economics, 76, 549-582.
- Foley, F. and R. Greenwood, 2009, The evolution of corporate ownership after the IPO: The impact of investor protection, Review of Financial Studies 23, 1231-1260.
- Frank, Murray Z. and Vidhan K. Goyal, 2003, Testing the pecking order theory of capital structure, Journal of Financial Economics 27, 217-248.
- Hendersen, B., Jagadeesh, N., and M. Weisbach, 2006, World markets for raising new capital, Journal of Financial Economics 82, 63-101.
- Hertzel, M. and Smith, R.L., 1993, Market discounts and shareholder gains for placing equity privately. Journal of Finance 48, 459–485.
- Jang, H., Kim, W., and Y. Ko, 2010, New Equity Issues in Emerging Economy: Do They Lead to Real Investments? Working Paper.
- Kim, Woojin, and Michael S. Weisbach, 2008, Motivations for Public Equity Offers: An International Perspective, Journal of Financial Economics, 87, 281-307.
- La Porta, Rafael, Florencio Lopez-de-Silanes, Andrei Shleifer, and Robert Vishny, 1997, Legal determinants of external finance, Journal of Finance 52, 1131–1150.
- La Porta, Rafael, Florencio Lopez-de-Silanes, Andrei Shleifer, and Robert W. Vishny, 1998, Law and Finance, Journal of Political Economy 106, 1113-1155.
- La Porta, Rafael, Florencio Lopez-de-Silanes, and Andrei Shleifer, 1999, Corporate ownership around the world, Journal of Finance 2, 471-517.
- La Porta, Rafael, Florencio Lopez-de-Silanes, Andrei Shleifer, and Robert W. Vishny, 2002, Investor protection and corporate valuation, Journal of Finance 57, 1147-1170.
- La Porta, Rafael, Florencio Lopez-de-Silanes, and Andrei Shleifer, 2006, What works in securities laws? Journal of Finance 61, 1-32.
- Leuz, Christian, Dhananjay Nanda, and Peter D. Wysocki, 2003, Earnings and investor protection: an international comparison, Journal of Financial Economics, 69, 505-527.

- Levine, Ross and Sarah Zervos, 1998, Stock Markets, Banks, and Economic Growth, American Economic Review 88, 537-558.
- McLean, R. David, 2010, Share Issuance and Cash Saving, Forthcoming, Journal of Financial Economics.
- McLean, R. David, Jeffrey Pontiff, and Akiko Watanabe, 2009, Share issuance and crosssectional returns: International evidence, Journal of Financial Economics 94, 1-17.
- Myers, Stuart, and Nicholas Majluf, 1984, Corporate financing and investment decisions when firms have information that investors do not have, Journal of Financial Economics 13, 187–221.
- Morck, Randall, Bernard Yeung, and Wayne Yu, 2000, The information content of stock markets: Why do emerging markets have synchronous stock price movements? Journal of Financial Economics 58, 215-260.
- Petersen, Mitchell, 2007, Estimating standard errors in finance panel data sets: comparing approaches, Forthcoming in the Review of Financial Studies 22, 435-480.
- Rajan, Raghuram and Luigi Zingales, 1995, What do we know about capital structure? Some evidence from international data, Journal of Finance, 50, 1421-1460.
- Rajan, Raghuram and Luigi Zingales, 2003, The great reversals: the politics of financial development in the twentieth century, Journal of Financial Economics, 60, 5-50.
- Reese, William A. Jr., and Michael S. Weisbach, 2002, Protection of minority shareholder interests, cross-listings in the United States, and subsequent equity offerings, Journal of Financial Economics 66, 65–104.
- Rossi, Stefano, and Volpin Paolo, 2004, Cross\_country determinants of mergers and acquisitions, Journal of Financial Economics, 74, 277-304.
- Stulz, Rene M., 1999, Globalization of equity markets and the cost of capital, Journal of Applied Corporate Finance 12, 8–25.
- White, Halbert, 1980, A Heteroskedasticity-Consistent Covariance Matrix Estimator and a Direct Test for Heteroskedasticity, Econometrica, 48, 817–838.
- Wruck, Karen Hopper, 1989, Equity Ownership Concentration and Firm Value: Evidence From Private Equity Financings, Journal of Financial Economics 23, 3-28.
- Wu, Yilin, 2004, The choice of equity-selling mechanisms, Journal of Financial Economics, 74, 93-119.

Zingales, L. 1995, What Determines the Value of Corporate Votes? Quarterly Journal of Economics 110, 1047–73.

# **Table 1: Share Issues over Time**

This table displays the frequencies (Panel A), frequencies as a percentage of firm-year observations (Panel B), and average amount in millions of real 2005 \$US (Panel C) of share issues around the world during the period 1990 through 2008. "Public" are seasoned equity offerings in the public market; "Mergers" are shares issued in mergers and acquisitions; "Private" are shares issued through private placements; "Rights" are shares issued through rights offerings to existing shareholders. "Total Issues" is the sum of the four issues. The dollar amount of a merger issue is calculated as the transaction value multiplied by the percentage of the transaction that is paid with equity. Proceeds from other share issues are measured as the product of the offer price and total number of shares issued.

Year	Public	Mergers	Private	Rights	Total Issues
1990	135	80	26	20	261
1991	404	129	34	49	616
1992	417	223	67	50	757
1993	724	249	89	98	1,160
1994	627	348	381	148	1,504
1995	650	443	415	139	1,647
1996	903	676	525	227	2,331
1997	892	821	561	217	2,491
1998	776	939	543	259	2,517
1999	1,288	1,041	859	280	3,468
2000	1,597	1,401	977	279	4,254
2001	1,063	991	879	281	3,214
2002	1,145	884	1,245	368	3,642
2003	1,172	730	1,683	314	3,899
2004	1,582	921	1,752	299	4,554
2005	1,618	1,084	1,881	319	4,902
2006	1,959	1,173	2,144	310	5,586
2007	2,531	1,376	2,252	482	6,641
2008	1,780	1,452	1,832	521	5,585
Total	21,263	14,961	18,145	4,660	59,029

**Panel A: Frequencies of Share Issues over Time** 

# Table 1 (Continued)

Year	Public	Mergers	Private	Rights	Total Issues
1990	0.63%	0.44%	0.06%	0.09%	1.19%
1991	1.36%	0.59%	0.10%	0.17%	2.12%
1992	1.29%	0.84%	0.17%	0.17%	2.39%
1993	2.11%	0.92%	0.25%	0.25%	3.35%
1994	1.87%	0.93%	0.38%	0.28%	3.23%
1995	2.08%	1.44%	0.48%	0.26%	3.99%
1996	2.32%	1.84%	0.50%	0.48%	4.84%
1997	2.21%	1.94%	0.68%	0.47%	4.89%
1998	1.68%	2.01%	0.61%	0.42%	4.38%
1999	2.65%	2.33%	0.86%	0.45%	5.78%
2000	3.28%	2.89%	1.18%	0.53%	7.08%
2001	2.04%	2.14%	1.43%	0.64%	5.73%
2002	2.60%	2.06%	2.25%	0.99%	7.23%
2003	2.64%	1.77%	3.09%	0.85%	7.51%
2004	3.76%	2.01%	3.39%	0.74%	9.00%
2005	3.86%	2.40%	3.60%	0.78%	9.48%
2006	4.41%	2.56%	3.96%	0.69%	10.31%
2007	5.55%	3.04%	4.32%	1.21%	12.15%
2008	3.89%	3.45%	3.78%	1.31%	10.40%
Total	2.93%	2.09%	2.01%	0.65%	6.88%

Panel B: Issue Frequency as a Percentage of Firm Observations over Time

# Table 1 (Continued)

Year	Public	Mergers	Private	Rights	Total Issues
1990	65.59	114.99	39.55	149.22	91.49
1991	72.76	110.93	83.80	165.56	103.75
1992	80.46	125.14	42.86	75.06	101.95
1993	166.19	137.60	111.15	42.47	171.21
1994	107.53	179.84	38.46	49.39	126.97
1995	138.04	199.03	33.86	77.37	156.58
1996	86.00	219.39	15.93	54.59	135.58
1997	111.42	266.52	30.85	84.00	183.79
1998	137.63	259.76	23.59	57.16	195.38
1999	133.39	504.03	38.52	56.50	294.05
2000	137.51	485.91	32.38	55.93	315.22
2001	104.10	310.65	19.06	92.27	189.91
2002	95.83	235.00	23.38	103.55	127.61
2003	78.32	286.09	33.95	72.42	134.80
2004	110.06	180.42	27.25	93.88	121.90
2005	101.88	288.44	24.10	49.16	147.76
2006	134.81	226.98	28.14	108.24	149.09
2007	131.07	781.22	36.04	65.87	413.24
2008	144.56	345.58	37.51	115.50	231.92
Total	118.93	340.83	30.57	79.33	203.79

Panel C: Mean Values of Share Issues (in millions of \$USD) over Time

# Table 2: Increase in SDC Issues Sample due to Factiva Search over Time

This table reports the percentage increase in our SDC sample due to our manual Factiva searches. The values reported in the cells are ratios; the numerator is the number of issues found in our Factiva searches, while the denominator is the number of issues from SDC. A higher value is associated with less complete SDC coverage. "Public" are seasoned equity offerings in the public market; "Mergers" are shares issued in mergers and acquisitions; "Private" are shares issued through private placements; "Rights" are shares issued through rights offerings to existing shareholders. "Total Issues" is the sum of the four issues.

Year	Public	Mergers	Private	Rights	Total Issues
1990	0.350	0.290	25.000	9.000	0.582
1991	0.307	0.206	2.400	0.441	0.339
1992	0.418	0.149	1.792	0.282	0.374
1993	0.281	0.097	2.296	0.324	0.299
1994	0.274	0.119	0.191	0.104	0.196
1995	0.204	0.124	0.189	0.198	0.177
1996	0.229	0.094	0.256	0.118	0.181
1997	0.368	0.151	0.524	0.167	0.298
1998	0.461	0.201	0.841	0.233	0.384
1999	0.505	0.241	0.515	0.191	0.389
2000	0.543	0.230	0.561	0.177	0.401
2001	0.370	0.354	0.348	0.120	0.333
2002	0.533	0.476	0.336	0.132	0.399
2003	0.414	0.396	0.314	0.113	0.338
2004	0.653	0.312	0.422	0.107	0.441
2005	0.853	0.417	0.528	0.152	0.558
2006	0.860	0.659	0.517	0.202	0.628
2007	0.617	1.018	0.498	0.164	0.595
2008	0.534	1.495	0.377	0.168	0.588
Total	0.511	0.401	0.442	0.167	0.429

# **Table 3: Share Issues Frequencies Summarized by Country**

This table reports share issues frequencies (Panel A) and frequencies as percentages of firm-year observations (Panel B) for 41 countries during the period 1990 to 2008. "Public" are seasoned equity offerings in the public market; "Mergers" are shares issued in mergers and acquisitions; "Private" are shares issued through private placements; "Rights" are shares issued through rights offerings to existing shareholders. "Total Issues" is the sum of the four issues.

# Table 3 (Continued)

		Pane	A: Issue Fr	equency		
Country	Public	Mergers	Private	Rights	Total Issues	% of Total
Argentina	16	7	0	53	76	0.13%
Australia	948	1,153	9,123	1,544	12,768	21.63%
Germany	319	75	35	89	518	0.88%
Belgium	49	20	25	18	112	0.19%
Brazil	112	45	8	35	200	0.34%
Colombia	10	18	3	19	50	0.08%
Chile	40	11	1	124	176	0.30%
Canada	2,683	916	1,221	65	4,885	8.28%
Denmark	93	45	14	34	186	0.32%
Spain	37	41	3	26	107	0.18%
Finland	71	106	12	18	207	0.35%
France	226	165	29	121	541	0.92%
Greece	37	18	5	14	74	0.13%
Hong Kong	786	166	292	322	1,566	2.65%
Indonesia	37	23	6	87	153	0.26%
India	612	338	779	188	1,917	3.25%
Ireland	97	29	47	11	184	0.31%
Israel	103	29	85	2	219	0.37%
Italy	146	48	29	48	271	0.46%
Japan	2,737	1,317	32	43	4,129	6.99%
Korea	436	239	474	239	1,388	2.35%
Mexico	160	43	67	41	311	0.53%
Malavsia	268	137	366	207	978	1.66%
Netherlands	208	121	65	13	407	0.69%
Norway	226	129	103	27	485	0.82%
New Zealand	82	79	109	51	321	0.54%
Austria	44	35	11	20	110	0.19%
Peru	10	12	5	3	30	0.05%
Philippines	66	63	75	61	265	0.45%
Pakistan	9	9	4	1	23	0.04%
Portugal	62	14	17	22	115	0.19%
South Africa	114	283	154	34	585	0.99%
Sweden	161	130	127	191	609	1.03%
Singapore	587	106	391	133	1 217	2.06%
Switzerland	95	14	26	38	173	0.29%
Taiwan	189	30	20	185	6/3	1.09%
Thailand	224	16	192	105	559	0.95%
Turkey	40	3	5	5	53	0.09%
I UIKCy I I K	1 327	800	5 451	189	2 767	4 69%
	7 783	8 115	3 510	167	19 570	
U.S. Vanazuelo	12	0,11J A	5,517	50	19,317 70	0 1 204
Total	1J 21 262	+ 14 061	J 18 145	JU 1 660	50.020	10004
rotai	21,203	14,901	10,143	4,000	39,029	100%

Panel	Δ۰	Issue	Free	mency	
I and	л.	199nc	ricy	ucity	

# Table 3 (Continued)

I unor L	D 11	<u>quency us u r</u>	Drivet		Trail
Ancontino	Public	Mergers	Private	Rights	1 otal Issues
Argentina	0.84%	0.30%	0.00%	1.31%	2.77%
Australia	5.54%	4.01%	18.88%	4.80%	24.44%
Germany	1.49%	0.44%	0.18%	0.47%	2.41%
Belgium	1.03%	0.55%	0.76%	0.45%	2.55%
Brazil	1.90%	0.80%	0.17%	0.43%	3.14%
Colombia	0.51%	1.53%	0.51%	1.87%	4.25%
Chile	0.96%	0.32%	0.04%	1.82%	2.86%
Canada	8.57%	3.08%	4.82%	0.27%	15.39%
Denmark	1.73%	1.07%	0.29%	0.81%	3.53%
Spain	0.81%	1.02%	0.06%	0.66%	2.46%
Finland	2.26%	2.83%	0.54%	0.54%	5.74%
France	1.11%	0.98%	0.16%	0.58%	2.70%
Greece	0.57%	0.38%	0.07%	0.31%	1.33%
Hong Kong	3.60%	0.99%	1.77%	1.65%	7.33%
Indonesia	0.42%	0.44%	0.07%	1.73%	2.64%
India	2.57%	1.56%	3.18%	0.76%	7.44%
Ireland	3.72%	1.82%	2.11%	0.70%	7.30%
Israel	2.44%	0.86%	2.74%	0.05%	6.00%
Italy	1.96%	0.72%	0.41%	0.70%	3.52%
Japan	3.57%	1.63%	0.05%	0.06%	5.10%
Korea	2.31%	1.75%	2.25%	1.52%	6.84%
Mexico	3.80%	1.49%	1.94%	1.45%	7.69%
Malaysia	1.46%	0.86%	2.38%	1.37%	5.83%
Netherlands	3.59%	2.41%	1.09%	0.21%	6.38%
Norway	4.31%	2.86%	2.41%	0.59%	9.04%
New Zealand	2.82%	2.99%	3.34%	1.78%	9.10%
Austria	1.63%	1.23%	0.46%	0.92%	3.52%
Peru	0.72%	0.89%	0.27%	0.18%	1.88%
Philippines	1.54%	1.68%	1.09%	1.33%	5.05%
Pakistan	0.39%	0.45%	0.17%	0.06%	1.01%
Portugal	1.89%	0.51%	0.66%	1.31%	3.93%
South Africa	0.83%	1.92%	1.37%	0.32%	4.22%
Sweden	1.79%	1.89%	1.62%	2.52%	7.08%
Singapore	4.75%	1.25%	3.56%	1.54%	10.34%
Switzerland	1.35%	0.47%	0.51%	0.59%	2.58%
Taiwan	1.12%	0.25%	1.33%	0.99%	3.48%
Thailand	2.52%	0.20%	2.15%	1.57%	5.99%
Turkev	0.85%	0.12%	0.12%	0.12%	1.16%
U.K	2.07%	1.44%	0.78%	0.39%	4.05%
U. S.	3.32%	3.41%	1.54%	0.06%	7.58%
Venezuela	0.77%	0.58%	0.77%	3.09%	5.03%
Total	2.93%	2.09%	2.01%	0.65%	6.88%

Panel B: Issue Frequency as a Percentage of Firm Observations

# Table 4: Increase in SDC Issue Sample due to Factiva Search by Country

This table reports the percentage increase in each country's SDC sample due to our manual Factiva searches. The values reported in the cells are ratios; the numerator is the number of issues found in our Factiva search, while the denominator is the number of issues from SDC. A higher value is associated with less complete SDC coverage. "Public" are seasoned equity offerings in the public market; "Mergers" are shares issued in mergers and acquisitions; "Private" are shares issued through private placements; "Rights" are shares issued through rights offerings to existing shareholders. "Total Issues" is the sum of the four issues. The sample period is 1990 through 2008. In some countries SDC has zero issues for a particular type of issue, but Factiva found one or more issue. In these instances we report "N/A" for not applicable, as we cannot report a ratio in these instances. Each country has more than one SDC issue in total, so we are able to report total issues statistics for each country.

Country	Public	Mergers	Private	Rights	Total Issues
Argentina	0.333	6.000	0.000	0.039	0.188
Australia	0.392	0.333	0.044	0.007	0.080
Germany	0.753	1.027	0.346	0.085	0.584
Belgium	0.324	0.538	0.667	0.000	0.349
Brazil	1.286	1.368	0.000	0.029	0.818
Colombia	1.500	0.500	0.500	0.118	0.429
Chile	0.818	4.500	0.000	0.000	0.181
Canada	0.142	0.221	5.938	2.421	0.483
Denmark	0.177	1.500	0.556	0.172	0.378
Spain	0.542	0.640	2.000	0.182	0.486
Finland	0.164	0.377	0.091	0.200	0.262
France	0.387	0.701	0.706	0.071	0.387
Greece	3.111	0.800	N/A	0.167	1.387
Hong Kong	0.351	0.509	0.884	0.283	0.426
Indonesia	2.364	N/A	0.500	0.176	0.719
India	3.192	7.450	7.954	0.528	3.841
Ireland	0.347	0.318	4.222	0.000	0.614
Israel	5.438	0.933	84.000	0.000	5.441
Italy	2.476	0.263	2.625	0.043	1.022
Japan	0.292	0.093	1.909	0.483	0.227
Korea	0.242	2.621	0.046	0.026	0.258
Mexico	2.478	5.143	15.750	12.667	4.183
Malaysia	4.057	0.181	0.887	0.035	0.737
Netherlands	0.763	4.500	1.241	0.182	1.261
Norway	0.263	1.481	1.512	0.080	0.633
New Zealand	1.485	1.548	0.730	0.041	0.824
Austria	0.630	34.000	1.200	0.000	1.075
Peru	2.333	5.000	4.000	2.000	3.286
Philippines	3.400	2.316	0.471	0.968	1.284
Pakistan	N/A	N/A	3.000	N/A	22.000
Portugal	4.636	13.000	4.667	0.100	2.286
South Africa	1.192	0.675	14.400	1.429	1.388
Sweden	0.695	0.368	8.071	0.240	0.701
Singapore	1.877	0.128	1.399	0.108	1.095
Switzerland	1.317	0.400	3.333	0.267	0.989
Taiwan	0.817	0.560	0.369	0.069	0.368
Thailand	6.467	0.333	0.864	0.270	1.282
Turkey	3.000	2.000	4.000	0.667	2.533
U.K.	0.166	0.493	1.148	2.048	0.422
U.S.	0.592	0.339	0.981	0.820	0.528
Venezuela	0.300	3.000	4.000	0.000	0.161
Total	0.511	0.401	0.442	0.167	0.429

 Table 4 (Continued)

# **Table 5: Investor Protection, Financial Development, and GDP Measures**

This table summarizes our investor protection, financial development, and GDP measures. Panel A lists the value of each measure for each country. Panel B reports the correlations between these measures and country-level measures of issuance frequency and amount. Common is equal to 1 if a country is of common law origin and zero if the country is of civil law origin. Disclosure is the arithmetic mean of six sub-indices. One of these sub-indices indicates whether or not new issues need to be accompanied by a prospectus. The five other sub-indices measure disclosure requirements within the prospectus regarding directors' and officers' compensations, controlling shareholders, insider ownership, irregular contracts, and any transactions between the issuer and its officers and directors. Liability is the arithmetic mean of three sub-indices. The sub-indices measure the ease with which an investor can pursue an issuer and its directors, the distributors, and the accountants in civil court if the investor suffers losses due to misleading statements in a prospectus. *Protect* is the first principal component of *Liability*, Disclosure, and an index of anti-director rights. Efficiency of the Judiciary measures the "efficiency and integrity of the legal environment as it affects business, particularly foreign firms." Efficiency of the Judiciary ranges in value from 0 to 10, with 10 meaning the best legal environment. Stock Market is the aggregate market capitalization of all publicly traded firms scaled by total GDP. Turnover is the total dollar value of stocks traded, scaled by total stock market capitalization, for the period 1996-2000. GDP is the average yearly value of per capita gross domestic product. Issue Frequency is the percentage of observations in a country that make at least one type of share issue. *Issue Amount* is the country-average of each firm-year observation's total issues scaled by lagged assets.

Table 5	(Continu	(ed
---------	----------	-----

			1	allel A				
Country	GDP	Common	Disclose	Liability	Protect	Eff_jud	StockMkt	Turnover
Argentina	9.21	0	0.50	0.22	0.48	6.00	58.08	20.74
Australia	10.30	1	0.75	0.66	0.78	10.00	101.97	66.02
Germany	10.29	0	0.42	0.00	0.00	9.00	54.69	143.54
Belgium	10.27	0	0.42	0.44	0.07	9.50	67.16	47.25
Brazil	8.98	0	0.25	0.33	0.44	5.75	38.35	48.12
Colombia	8.81	0	0.42	0.11	0.35	7.25	14.27	9.59
Chile	9.21	0	0.58	0.33	0.61	7.25	89.70	11.98
Canada	10.33	1	0.92	1.00	0.96	9.25	106.18	67.79
Denmark	10.31	0	0.58	0.55	0.36	10.00	58.60	69.75
Spain	10.08	0	0.50	0.66	0.55	6.25	79.91	164.08
Finland	10.17	0	0.50	0.66	0.47	10.00	177.11	95.72
France	10.21	0	0.75	0.22	0.47	8.00	89.49	92.28
Greece	9.94	0	0.33	0.50	0.32	7.00	91.38	49.52
Hong Kong	10.31	1	0.92	0.66	0.85	10.00	360.98	58.37
Indonesia	7.95	0	0.50	0.66	0.51	2.50	24.70	52.48
India	7.43	1	0.92	0.66	0.77	8.00	33.80	84.22
Ireland	10.23	1	0.67	0.44	0.48	8.75	67.65	59.28
Israel	9.97	1	0.67	0.66	0.59	10.00	53.04	47.89
Italy	10.19	0	0.67	0.22	0.20	6.75	52.77	113.35
Japan	10.26	0	0.75	0.66	0.42	10.00	69.17	79.85
Korea	9.79	0	0.75	0.66	0.36	6.00	54.14	190.97
Mexico	9.35	0	0.58	0.11	0.10	6.00	21.87	34.51
Malaysia	9.19	1	0.92	0.66	0.73	9.00	148.42	40.90
Netherlands	10.36	0	0.50	0.89	0.54	10.00	131.74	117.82
Norway	10.63	0	0.58	0.39	0.44	10.00	39.69	94.74
New Zealand	9.98	1	0.67	0.44	0.46	10.00	40.10	40.15
Austria	10.31	0	0.25	0.11	0.10	9.50	16.39	50.98
Peru	8.62	0	0.33	0.66	0.66	6.75	22.85	18.07
Philippines	7.86	0	0.83	1.00	0.81	4.75	48.00	25.10
Pakistan	7.58	1	0.58	0.39	0.63	5.00	14.32	177.73
Portugal	9.84	0	0.42	0.66	0.57	5.5	46.24	61.47
South Africa	8.97	1	0.83	0.66	0.60	6.00	155.77	38.64
Sweden	10.23	0	0.58	0.28	0.39	10.00	112.27	98.33
Singapore	10.45	1	1.00	0.66	0.77	10.00	164.75	60.53
Switzerland	10.44	0	0.67	0.44	0.30	10.00	248.96	98.58
Taiwan	9.88	0	0.75	0.66	0.55	6.75	101.89	314.74
Thailand	8.64	1	0.92	0.22	0.37	3.25	44.79	83.65
Turkey	9.14	0	0.50	0.22	0.34	4.00	35.30	134.57
U.K.	10.25	1	0.83	0.66	0.78	10.00	157.70	105.41
U.S.	10.52	1	1.00	1.00	1.00	10.00	142.14	145.30
Venezuela	9.21	0	0.17	0.22	0.22	6.5	5.51	16.77
Mean	9 65	0 34	0.63	0.50	0.50	7 81	83 95	75 40
Std. Dev.	0.86	0.48	0.03	0.26	0.24	2.19	69.65	45 98
	0.00	0.10	J	J	·· ·		02.00	

Panel A

I able 5 (Comunicu)	Table	e 5	(Continu	ied)
---------------------	-------	-----	----------	------

					Panel B					
	Issue Freq	Issue/ Assets	GDP	Common	Disclose	Liability	Protect	Eff_jud	StockMkt	Turnover
Issue Freq	1									
Issue/Assets	0.918	1								
GDP	0.279	0.285	1							
Common	0.486	0.456	-0.053	1						
Disclose	0.443	0.291	0.047	0.698	1					
Liability	0.339	0.373	0.032	0.366	0.498	1				
Protect	0.422	0.437	-0.137	0.626	0.643	0.773	1			
Eff_jud	0.393	0.344	0.730	0.229	0.217	0.199	0.132	1		
StockMkt	0.174	0.124	0.426	0.318	0.479	0.370	0.399	0.476	1	
Turnover	-0.064	-0.067	0.199	0.023	0.186	0.090	-0.017	0.024	0.086	1

### **Table 6: Total Issues Country-Level Regressions**

This table reports results from country-level regressions The dependent variable is the percentage of observations in a country that make at least one type of share issue. The independent variables include: Common is equal to 1 if a country is of common law origin and zero if the country is of civil law origin. Disclosure is the arithmetic mean of six sub-indices. One of these sub-indices indicates whether or not new issues need to be accompanied by a prospectus. The five other sub-indices measure disclosure requirements within the prospectus regarding directors' and officers' compensations, controlling shareholders, insider ownership, irregular contracts, and any transactions between the issuer and its officers and directors. Liability is the arithmetic mean of three sub-indices. The subindices measure the ease with which an investor can pursue an issuer and its directors, the distributors, and the accountants in civil court if the investor suffers losses due to misleading statements in a prospectus. Protect is the first principal component of Liability, Disclosure, and an index of anti-director rights. Efficiency of the Judiciary measures the "efficiency and integrity of the legal environment as it affects business, particularly foreign firms." Efficiency of the Judiciary ranges in value from 0 to 10, with 10 meaning the best legal environment. Stock Market is the aggregate market capitalization of all publicly traded firms scaled by total GDP. Turnover is the total dollar value of stocks traded, scaled by total stock market capitalization, for the period 1996-2000. GDP is the average yearly value of per capita gross domestic product. White (1980) heteroskedasticity-consistent t-statistics are reported in the parentheses. \* Significant at 10%; \*\* Significant at 5%; \*\*\* Significant at 1%.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
GDP	0.013*	0.015**	0.012**	0.013**	0.016**	-0.001	0.012	0.015*
	(1.94)	(2.24)	(2.31)	(2.07)	(2.52)	(0.23)	(1.62)	(1.99)
Common		0.044***						
		(2.98)						
Disclose			0.081***					
			(3.76)					
Liability				0.052**				
				(2.53)				
Protect					0.082**			
					(2.61)			
Eff_jud						0.008**		
						(2.68)		
StockMkt							0.004	
							(0.69)	
Turnover								-0.011
								(1.76)
Constant	-0.074	-0.103	-0.115**	-0.095	-0.144*	0.005	-0.064	-0.078
	(1.17)	(1.61)	(2.06)	(1.54)	(2.00)	(0.11)	(0.97)	(1.20)
Observations	41	41	41	41	41	41	41	41
R-squared	0.08	0.33	0.25	0.18	0.29	0.16	0.08	0.10

# **Table 7: Aggregate Issues Firm-level Regressions**

This table reports the results from firm-level Probit regressions. The dependent variable is equal to 1 if the firm issues shares in a seasoned offering, and zero otherwise. The control variables include: Assets is the log of book value of total assets at the beginning of the issuing year; MB is the log of market-to-book ratio of assets at the beginning of the issuing year; CF is operating cash flow scaled by book value of total assets at the beginning of the issuing year; Leverage is the ratio of total debt to assets at the beginning of the issuing year; PPE is property plant and equipment scaled by total assets. Common is equal to 1 if a country is of common law origin and zero if the country is of civil law origin. Disclosure is the arithmetic mean of six sub-indices. One of these sub-indices indicates whether or not new issues need to be accompanied by a prospectus. The five other sub-indices measure disclosure requirements within the prospectus regarding directors' and officers' compensations, controlling shareholders, insider ownership, irregular contracts, and any transactions between the issuer and its officers and directors. Liability is the arithmetic mean of three sub-indices. The sub-indices measure the ease with which an investor can pursue an issuer and its directors, the distributors, and the accountants in civil court if the investor suffers losses due to misleading statements in a prospectus. Protect is the principal component of Liability, Disclosure, and an index of anti-director rights. Efficiency of the Judiciary measures the "efficiency and integrity of the legal environment as it affects business, particularly foreign firms." Efficiency of the Judiciary ranges in value from 0 to 10, with 10 meaning the best legal environment. Stock Market is the aggregate market capitalization of all publicly traded firms scaled by total GDP. Turnover is the total dollar value of stocks traded, scaled by total stock market capitalization, for the period 1996-2000. GDP is the average yearly value of per capita gross domestic product. All regressions include industry and year fixed effects. The coefficients reflect marginal probabilities. Robust t-statistics are computed by clustering on the country and are reported in parentheses. \* Significant at 10%; \*\* Significant at 5%; \*\*\* Significant at 1%.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
GDP	0.011	0.008*	0.006	0.004	0.004	-0.002	0.009	0.012
	(1.32)	(1.65)	(1.08)	(0.62)	(0.89)	(0.32)	(0.97)	(1.36)
Common		0.034***						
		(4.97)						
Disclose			0.074***					
			(2.71)					
Liability				0.048***				
				(3.13)				
Investor_pr					0.055***			
					(5.75)			
Eff_Jud						0.008*		
						(1.91)		
StockMkt							0.007	
							(1.00)	
Turnover								-0.007
								(0.77)
Assets	0.001	0.002	0.001	0.001	0.001	0.001	0.001	0.001
	(0.39)	(1.02)	(0.50)	(0.37)	(0.63)	(0.48)	(0.44)	(0.42)
MB	0.023***	0.022***	0.024***	0.022***	0.021***	0.023***	0.023***	0.023***
	(3.01)	(3.70)	(4.50)	(3.39)	(3.18)	(3.40)	(3.09)	(2.95)
Leverage	0.032***	0.034***	0.033***	0.032***	0.033***	0.033***	0.033***	0.033***
	(5.32)	(5.44)	(5.12)	(5.18)	(5.07)	(5.62)	(5.10)	(5.86)
PPE	0.014**	0.010*	0.013**	0.012**	0.011*	0.014**	0.013*	0.014**
	(2.06)	(1.76)	(2.09)	(1.96)	(1.87)	(2.13)	(1.93)	(2.07)
CF	-0.073***	-0.068***	-0.069***	-0.069***	-0.068***	-0.071***	-0.073***	-0.073***
	(5.15)	(5.35)	(4.98)	(5.04)	(5.24)	(5.24)	(5.16)	(5.21)
Observations	292,685	292,685	292,685	292,685	292,685	292,685	292,685	292,685
Pseudo R <sup>2</sup>	0.14	0.15	0.14	0.14	0.14	0.14	0.14	0.14

# Table 7 (Continued)

# **Table 8: Public Offerings Firm-Level Regressions**

This table reports the results from firm-level Probit regressions. The dependent variable is equal to 1 if the firm issues shares in a public offering, and zero otherwise. The control variables include: Assets is the log of book value of total assets at the beginning of the issuing year; MB is the log of market-to-book ratio of assets at the beginning of the issuing year; CF is operating cash flow scaled by book value of total assets at the beginning of the issuing year; Leverage is the ratio of total debt to assets at the beginning of the issuing year; PPE is property plant and equipment scaled by total assets. Common is equal to 1 if a country is of common law origin and zero if the country is of civil law origin. Disclosure is the arithmetic mean of six sub-indices. One of these sub-indices indicates whether or not new issues need to be accompanied by a prospectus. The five other sub-indices measure disclosure requirements within the prospectus regarding directors' and officers' compensations, controlling shareholders, insider ownership, irregular contracts, and any transactions between the issuer and its officers and directors. Liability is the arithmetic mean of three sub-indices. The sub-indices measure the ease with which an investor can pursue an issuer and its directors, the distributors, and the accountants in civil court if the investor suffers losses due to misleading statements in a prospectus. Protect is the principal component of Liability, Disclosure, and an index of anti-director rights. Efficiency of the Judiciary measures the "efficiency and integrity of the legal environment as it affects business, particularly foreign firms." Efficiency of the Judiciary ranges in value from 0 to 10, with 10 meaning the best legal environment. Stock Market is the aggregate market capitalization of all publicly traded firms scaled by total GDP. Turnover is the total dollar value of stocks traded, scaled by total stock market capitalization, for the period 1996-2000. GDP is the average yearly value of per capita gross domestic product. The coefficients reflect marginal probabilities. All regressions include industry and year fixed effects. Robust t-statistics are computed by clustering on the country and are reported in parentheses. \* Significant at 10%; \*\* Significant at 5%; \*\*\* Significant at 1%.

Table	8	(Continued)
-------	---	-------------

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
GDP	0.010**	0.008***	0.007***	0.006**	0.007**	0.004	0.009**	0.010**
	(2.25)	(2.59)	(2.62)	(2.05)	(2.28)	(1.08)	(2.04)	(2.28)
Common		0.008**						
		(2.00)						
Disclose			0.029***					
			(3.63)					
Liability				0.022***				
				(3.55)				
Investor_pr					0.015**			
					(2.48)			
Eff_Jud						0.003*		
						(1.92)		
StockMkt							0.002	
							(0.85)	
Turnover								-0.003
								(1.46)
Assets	0.001*	0.001**	0.001**	0.001*	0.001**	0.001*	0.001**	0.001*
	(1.91)	(2.29)	(2.29)	(1.96)	(2.26)	(1.96)	(1.99)	(1.88)
MB	0.010***	0.010***	0.011***	0.010***	0.010***	0.011***	0.010***	0.010***
	(3.93)	(4.37)	(6.65)	(4.53)	(3.98)	(4.45)	(4.01)	(3.82)
Leverage	0.019***	0.020***	0.020***	0.019***	0.020***	0.019***	0.020***	0.020***
	(5.15)	(5.73)	(5.54)	(5.48)	(5.47)	(5.25)	(5.24)	(5.35)
PPE	0.005	0.004	0.005	0.004	0.004	0.005	0.005	0.005
	(1.43)	(1.29)	(1.42)	(1.35)	(1.31)	(1.51)	(1.38)	(1.47)
CF	-0.013**	-0.012*	-0.012*	-0.012*	-0.012**	-0.013*	-0.013**	-0.013**
	(1.96)	(1.95)	(1.91)	(1.94)	(1.96)	(1.93)	(1.96)	(1.96)
Observations	292,646	292,646	292,646	292,646	292,646	292,646	292,646	292,646
Pseudo R <sup>2</sup>	0.10	0.10	0.10	0.11	0.10	0.10	0.10	0.10

## **Table 9: Stock-Financed Mergers Firm-Level Regressions**

This table reports the results from firm-level Probit regressions. The dependent variable is equal to 1 if the firm issues shares in a merger, and zero otherwise. The control variables include: Assets is the log of book value of total assets at the beginning of the issuing year; MB is the log of market-to-book ratio of assets at the beginning of the issuing year; CF is operating cash flow scaled by book value of total assets at the beginning of the issuing year; Leverage is the ratio of total debt to assets at the beginning of the issuing year; PPE is property plant and equipment scaled by total assets. Common is equal to 1 if a country is of common law origin and zero if the country is of civil law origin. Disclosure is the arithmetic mean of six sub-indices. One of these sub-indices indicates whether or not new issues need to be accompanied by a prospectus. The five other sub-indices measure disclosure requirements within the prospectus regarding directors' and officers' compensations, controlling shareholders, insider ownership, irregular contracts, and any transactions between the issuer and its officers and directors. Liability is the arithmetic mean of three sub-indices. The sub-indices measure the ease with which an investor can pursue an issuer and its directors, the distributors, and the accountants in civil court if the investor suffers losses due to misleading statements in a prospectus. Protect is the first principal component of Liability, Disclosure, and an index of antidirector rights. Efficiency of the Judiciary measures the "efficiency and integrity of the legal environment as it affects business, particularly foreign firms." Efficiency of the Judiciary ranges in value from 0 to 10, with 10 meaning the best legal environment. Stock Market is the aggregate market capitalization of all publicly traded firms scaled by total GDP. Turnover is the total dollar value of stocks traded, scaled by total stock market capitalization, for the period 1996-2000. GDP is the average yearly value of per capita gross domestic product. The coefficients reflect marginal probabilities. All regressions include industry and year fixed effects. Robust t-statistics are computed by clustering on the country and are reported in parentheses. \* Significant at 10%; \*\* Significant at 5%; \*\*\* Significant at 1%.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
GDP	0.009*	0.006**	0.005**	0.003	0.004**	0.002	0.008	0.009*
	(1.74)	(2.46)	(2.10)	(1.43)	(2.06)	(0.59)	(1.63)	(1.74)
Common		0.012***						
		(4.99)						
Disclose			0.034***					
			(4.85)					
Liability				0.027***				
				(6.67)				
Investor_pr					0.024***			
					(9.10)			
Eff_Jud						0.004**		
						(2.27)		
StockMkt							0.001	
							(0.40)	
Turnover								-0.000
								(0.03)
Assets	0.003***	0.003***	0.003***	0.003***	0.003***	0.003***	0.003***	0.003***
	(5.09)	(4.84)	(4.46)	(4.31)	(4.39)	(5.73)	(4.81)	(5.11)
MB	0.008***	0.007***	0.008***	0.007***	0.007***	0.008***	0.008***	0.008***
	(3.91)	(4.64)	(7.56)	(4.95)	(4.01)	(4.57)	(3.97)	(3.93)
Leverage	0.004*	0.005**	0.004***	0.004***	0.004**	0.004**	0.004**	0.004*
	(1.93)	(2.54)	(2.64)	(2.62)	(2.51)	(2.15)	(2.06)	(1.96)
PPE	-0.000	-0.001	-0.000	-0.001	-0.001	0.000	-0.000	-0.000
	(0.06)	(0.64)	(0.24)	(0.62)	(0.58)	(0.04)	(0.11)	(0.06)
CF	-0.007***	-0.006***	-0.006***	-0.005***	-0.006***	-0.007***	-0.007***	-0.007***
	(3.37)	(3.15)	(2.95)	(2.70)	(2.88)	(3.30)	(3.34)	(3.37)
Observations	279,137	279,137	279,137	279,137	279,137	279,137	279,137	279,137
Pseudo R <sup>2</sup>	0.09	0.10	0.10	0.11	0.11	0.10	0.09	0.09

# **Table 10: Private Placements Firm-Level Regressions**

This table reports the results from firm-level Probit regressions. The dependent variable is equal to 1 if the firm issues shares in a private placement, and zero otherwise. The control variables include: Assets is the log of book value of total assets at the beginning of the issuing year; MB is the log of market-to-book ratio of assets at the beginning of the issuing year; CF is operating cash flow scaled by book value of total assets at the beginning of the issuing year; Leverage is the ratio of total debt to assets at the beginning of the issuing year; PPE is property plant and equipment scaled by total assets. Common is equal to 1 if a country is of common law origin and zero if the country is of civil law origin. Disclosure is the arithmetic mean of six sub-indices. One of these sub-indices indicates whether or not new issues need to be accompanied by a prospectus. The five other sub-indices measure disclosure requirements within the prospectus regarding directors' and officers' compensations, controlling shareholders, insider ownership, irregular contracts, and any transactions between the issuer and its officers and directors. Liability is the arithmetic mean of three sub-indices. The sub-indices measure the ease with which an investor can pursue an issuer and its directors, the distributors, and the accountants in civil court if the investor suffers losses due to misleading statements in a prospectus. *Protect* is the first principal component of *Liability*, Disclosure, and an index of anti-director rights. Efficiency of the Judiciary measures the "efficiency and integrity of the legal environment as it affects business, particularly foreign firms." Efficiency of the Judiciary ranges in value from 0 to 10, with 10 meaning the best legal environment. Stock Market is the aggregate market capitalization of all publicly traded firms scaled by total GDP. *Turnover* is the total dollar value of stocks traded, scaled by total stock market capitalization, for the period 1996-2000. GDP is the average yearly value of per capita gross domestic product. All regressions include industry and year fixed effects. The coefficients reflect marginal probabilities. Robust t-statistics are computed by clustering on the country and are reported in parentheses. \* Significant at 10%; \*\* Significant at 5%; \*\*\* Significant at 1%.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
GDP	-0.001	-0.001	-0.001	-0.001	-0.002	-0.003***	-0.001	-0.001
	(0.44)	(0.70)	(0.54)	(0.49)	(1.04)	(3.56)	(0.46)	(0.29)
Common		0.010***						
		(3.13)						
Disclose			0.007					
			(0.61)					
Liability				0.003				
				(0.43)				
Investor_pr					0.012**			
					(1.99)			
Eff_Jud						0.002		
						(1.34)		
StockMkt							0.001	
							(0.38)	
Turnover								-0.002
								(0.62)
Assets	-0.002*	-0.001	-0.002	-0.002*	-0.001*	-0.002*	-0.002	-0.002*
	(1.68)	(1.58)	(1.63)	(1.71)	(1.66)	(1.71)	(1.62)	(1.71)
MB	0.001	0.001	0.002	0.001	0.001	0.001	0.001	0.001
	(1.29)	(1.30)	(1.33)	(1.34)	(1.16)	(1.46)	(1.30)	(1.30)
Leverage	0.003*	0.003*	0.003*	0.003*	0.003*	0.003*	0.003*	0.003**
	(1.85)	(1.86)	(1.77)	(1.80)	(1.70)	(1.87)	(1.81)	(2.15)
PPE	0.005***	0.003***	0.004***	0.004***	0.004***	0.005***	0.005***	0.005***
	(3.73)	(3.37)	(3.38)	(3.26)	(3.28)	(3.86)	(3.68)	(3.73)
CF	-0.020***	-0.016***	-0.019***	-0.019***	-0.018***	-0.019***	-0.020***	-0.020***
	(10.13)	(11.17)	(10.70)	(10.91)	(10.75)	(10.32)	(10.24)	(10.40)
Observations	292,105	292,105	292,105	292,105	292,105	292,105	292,105	292,105
Pseudo R <sup>2</sup>	0.23	0.24	0.23	0.23	0.23	0.23	0.23	0.23

# Table 10 (Continued)

# **Table 11: Rights Offerings Firm-Level Regressions**

This table reports the results from firm-level Probit regressions. The dependent variable is equal to 1 if the firm issues shares in a rights offering, and zero otherwise. The control variables include: Assets is the log of book value of total assets at the beginning of the issuing year; MB is the log of market-to-book ratio of assets at the beginning of the issuing year; CF is operating cash flow scaled by book value of total assets at the beginning of the issuing year; Leverage is the ratio of total debt to assets at the beginning of the issuing year; PPE is property plant and equipment scaled by total assets. Common is equal to 1 if a country is of common law origin and zero if the country is of civil law origin. Disclosure is the arithmetic mean of six sub-indices. One of these sub-indices indicates whether or not new issues need to be accompanied by a prospectus. The five other sub-indices measure disclosure requirements within the prospectus regarding directors' and officers' compensations, controlling shareholders, insider ownership, irregular contracts, and any transactions between the issuer and its officers and directors. Liability is the arithmetic mean of three sub-indices. The sub-indices measure the ease with which an investor can pursue an issuer and its directors, the distributors, and the accountants in civil court if the investor suffers losses due to misleading statements in a prospectus. Protect is the first principal component of Liability, Disclosure, and an index of antidirector rights. Efficiency of the Judiciary measures the "efficiency and integrity of the legal environment as it affects business, particularly foreign firms." Efficiency of the Judiciary ranges in value from 0 to 10, with 10 meaning the best legal environment. Stock Market is the aggregate market capitalization of all publicly traded firms scaled by total GDP. Turnover is the total dollar value of stocks traded, scaled by total stock market capitalization, for the period 1996-2000. GDP is the average yearly value of per capita gross domestic product. The coefficients reflect marginal probabilities. All regressions include industry and year fixed effects. Robust t-statistics are computed by clustering on the country and are reported in parentheses. \* Significant at 10%; \*\* Significant at 5%; \*\*\* Significant at 1%.

# Table 11 (Continued)

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
GDP	-0.002	-0.002	-0.002	-0.001	-0.002	-0.001	-0.002	-0.002
	(1.50)	(1.50)	(1.58)	(1.10)	(1.53)	(1.17)	(1.52)	(1.36)
Common		-0.000						
		(0.13)						
Disclose			-0.010**					
			(2.08)					
Liability				-0.009***				
				(3.28)				
Investor_pr					-0.005			
					(1.42)			
Eff_Jud						-0.000		
						(0.88)		
StockMkt							0.001	
							(0.66)	
Turnover								-0.002
								(0.87)
Assets	-0.001	-0.001	-0.001	-0.001	-0.001	-0.001	-0.001	-0.001
	(1.32)	(1.56)	(1.51)	(1.46)	(1.59)	(1.37)	(1.21)	(1.34)
MB	-0.001	-0.001	-0.001	-0.001	-0.001	-0.001	-0.001	-0.001
	(1.13)	(1.15)	(1.53)	(1.27)	(1.03)	(1.18)	(1.06)	(1.17)
Leverage	0.004***	0.004***	0.003***	0.003***	0.003***	0.004***	0.004**	0.004***
	(2.62)	(2.63)	(2.87)	(2.89)	(2.72)	(2.69)	(2.54)	(3.29)
PPE	0.002	0.002*	0.002	0.002**	0.002*	0.002	0.002	0.002
	(1.55)	(1.70)	(1.61)	(2.15)	(1.92)	(1.53)	(1.46)	(1.56)
CF	-0.008***	-0.008***	-0.008***	-0.008***	-0.008***	-0.008***	-0.008***	-0.008**
	(5.96)	(6.49)	(7.09)	(7.52)	(7.21)	(6.06)	(5.96)	(5.93)
Observations	291,795	291,795	291,795	291,795	291,795	291,795	291,795	291,795
Pseudo R <sup>2</sup>	0.09	0.09	0.11	0.12	0.10	0.09	0.09	0.10

# Table 12: Rights Offerings vs. Other Types of Issues Firm-Level Regressions

This table reports OLS regression tests that measure the use of rights offerings vs. other issue types. The dependent variable is the amount of capital raised in rights offerings scaled by the sum of capital raised in public offerings, rights, and private placements. The control variables include: Assets is the log of book value of total assets at the beginning of the issuing year; *MB* is the log of market-to-book ratio of assets at the beginning of the issuing year; CF is operating cash flow scaled by book value of total assets at the beginning of the issuing year; Leverage is the ratio of total debt to assets at the beginning of the issuing year; PPE is property plant and equipment scaled by total assets. Common is equal to 1 if a country is of common law origin and zero if the country is of civil law origin. Disclosure is the arithmetic mean of six sub-indices. One of these sub-indices indicates whether or not new issues need to be accompanied by a prospectus. The five other sub-indices measure disclosure requirements within the prospectus regarding directors' and officers' compensations, controlling shareholders, insider ownership, irregular contracts, and any transactions between the issuer and its officers and directors. Liability is the arithmetic mean of three sub-indices. The sub-indices measure the ease with which an investor can pursue an issuer and its directors, the distributors, and the accountants in civil court if the investor suffers losses due to misleading statements in a prospectus. Protect is the first principal component of Liability, Disclosure, and an index of anti-director rights. Efficiency of the Judiciary measures the "efficiency and integrity of the legal environment as it affects business, particularly foreign firms." Efficiency of the Judiciary ranges in value from 0 to 10, with 10 meaning the best legal environment. Stock Market is the aggregate market capitalization of all publicly traded firms scaled by total GDP. Turnover is the total dollar value of stocks traded, scaled by total stock market capitalization, for the period 1996-2000. GDP is the average yearly value of per capita gross domestic product. All regressions include industry and year fixed effects. Robust t-statistics are computed by clustering on the country and are reported in parentheses. \* Significant at 10%; \*\* Significant at 5%; \*\*\* Significant at 1%.

# Table 12 (Continued)

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
GDP	-0.072*	-0.067*	-0.058*	-0.030	-0.048	-0.018	-0.080*	-0.070
	(1.68)	(1.91)	(1.98)	(1.07)	(1.54)	(0.53)	(1.79)	(1.66)
Common		-0.089						
		(1.36)						
Disclose			-0.395***					
			(4.43)					
Liability				-0.348***				
				(7.44)				
Investor_pr					-0.231**			
					(2.43)			
Eff_Jud						-0.038***		
						(3.10)		
StockMkt							0.022	
							(0.84)	
Turnover								-0.014
								(0.37)
Assets	-0.008**	-0.011***	-0.008**	-0.005*	-0.010**	-0.010***	-0.008**	-0.008**
	(2.53)	(3.05)	(2.16)	(1.84)	(2.66)	(3.12)	(2.32)	(2.11)
MB	-0.045***	-0.043***	-0.041***	-0.031***	-0.036***	-0.045***	-0.044***	-0.044***
	(4.57)	(4.18)	(3.73)	(3.50)	(3.63)	(5.07)	(4.67)	(4.58)
Leverage	0.003	-0.005	-0.000	-0.004	-0.007	0.000	0.006	0.005
	(0.18)	(0.28)	(0.03)	(0.26)	(0.45)	(0.01)	(0.30)	(0.26)
PPE	0.004	0.011	0.010	0.019	0.015	0.001	0.003	0.004
	(0.23)	(0.79)	(0.77)	(1.57)	(1.14)	(0.04)	(0.21)	(0.24)
CF	-0.023	-0.024*	-0.025*	-0.031**	-0.027**	-0.020	-0.024	-0.023*
	(1.66)	(1.78)	(1.88)	(2.42)	(2.15)	(1.52)	(1.63)	(1.71)
Constant	0.777*	0.812**	0.874***	0.695**	0.702**	0.658**	0.835**	0.772*
	(1.99)	(2.65)	(3.31)	(2.55)	(2.55)	(2.21)	(2.05)	(1.99)
Observations	18,167	18,167	18,167	18,167	18,167	18,167	18,167	18,167
Pseudo R <sup>2</sup>	0.06	0.07	0.10	0.12	0.09	0.08	0.06	0.06